Author's response to reviews

Title: Haplotype frequencies in a sub-region of chromosome 19q13.3, related to risk and prognosis of cancer, differ dramatically between ethnic groups

Authors:

Mikkel H Schierup (mheide@daimi.au.dk)
Thomas Mailund (mailund@birc.au.dk)
Heng Li (liheng@genomics.org.cn)
Jun Wang (wangj@genomics.org.cn)
Anne Tjønneland (annet@cancer.dk)
Ulla Vogel (ulbvo@food.dtu.dk)
Lars Bolund (bolund@humgen.au.dk)
Bjørn A Nexø (nexo@humgen.au.dk)

Version: 6 Date: 17 February 2009

Author's response to reviews:

We have now also added a paragraph about "Competing Interests"

Yours sincerely

Bjorn Nexø